

Delft University of Technology

First Application of Artificial Neural Networks to Estimate 21st Century Greenland Ice Sheet Surface Melt

Sellevold, Raymond; Vizcaino, Miren

DOI 10.1029/2021GL092449

Publication date 2021 **Document Version** Final published version

Published in **Geophysical Research Letters**

Citation (APA) Sellevold, R., & Vizcaino, M. (2021). First Application of Artificial Neural Networks to Estimate 21st Century Greenland Ice Sheet Surface Melt. Geophysical Research Letters, 48(16), Article e2021GL092449. https://doi.org/10.1029/2021GL092449

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

1 of 16



Geophysical Research Letters

RESEARCH LETTER

10.1029/2021GL092449

Special Section:

Community Earth System Model version 2 (CESM2) Special Collection

Key Points:

- We explore connections between atmospheric variables from global climate simulations and Greenland melt using neural networks
- The neural networks' melt predictions compare well to those calculated by a global and a regional climate model
- Climate model spread is the main source of uncertainty in 21st century Greenland melt projections

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to: R. Sellevold, R.Sellevold-1@tudelft.nl

Citation:

Sellevold, R., & Vizcaino, M. (2021). First application of artificial neural networks to estimate 21st century Greenland ice sheet surface melt. *Geophysical Research Letters*, 48, e2021GL092449. https://doi. org/10.1029/2021GL092449

Received 7 JAN 2021 Accepted 26 JUL 2021

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

First Application of Artificial Neural Networks to Estimate 21st Century Greenland Ice Sheet Surface Melt

Raymond Sellevold¹ and Miren Vizcaino¹

¹Geoscience and Remote Sensing, Delft University of Technology, Delft, The Netherlands

Abstract Future Greenland ice sheet (GrIS) melt projections are limited by the lack of explicit melt calculations within most global climate models and the high computational cost of dynamical downscaling with regional climate models (RCMs). Here, we train artificial neural networks (ANNs) to obtain relationships between quantities consistently available from global climate model simulations and annually integrated GrIS surface melt. To this end, we train the ANNs with model output from the Community Earth System Model 2.1 (CESM2), which features an interactive surface melt calculation based on a downscaled surface energy balance. We find that ANNs compare well with an independent CESM2 simulation and RCM simulations forced by a CMIP6 subset. The ANNs estimate a melt increase for 2,081–2,100 ranging from 414 ± 275 Gt yr⁻¹ (SSP1-2.6) to 1,378 ± 555 Gt yr⁻¹ (SSP5-8.5) for the full CMIP6 suite. The primary source of uncertainty throughout the 21st century is the spread of climate model sensitivity.

Plain Language Summary The accelerated melt of the Greenland ice sheet (GrIS) surface is currently a large contributor to sea level rise. Most global climate models used for climate projections do not calculate GrIS melt due to limited snow physics and model resolution. Here, we use for the first-time artificial intelligence to calculate GrIS melt from atmospheric variables of all global climate models contributing to the next Intergovernmental Panel on Climate Change report and for a representative range of greenhouse gas emission scenarios. To this end, we train artificial intelligence with one of the few global climate models that explicitly calculates GrIS melt. We find that limiting emissions is effective in restricting future GrIS surface melt increase.

1. Introduction

Greenland is losing mass at an accelerating rate since the 1990s (Bamber et al., 2018; Mouginot et al., 2019; Shepherd et al., 2020) in response to global warming. This is alarming, as the freshwater stored on the Greenland ice sheet (GrIS) has the potential to raise the global mean sea level by 7.34 m if fully melted (Bamber et al., 2018). The current mass loss of the GrIS is the result of increased ice discharge and decreased surface mass balance (SMB), with the latter being the dominant contributor (Fettweis et al., 2017; van den Broeke et al., 2016), and the cause of mass loss acceleration (Enderlin et al., 2014; Shepherd et al., 2020). The main contributor to this contemporary GrIS surface mass loss is increased surface melt (Fettweis et al., 2017; Noël et al., 2020). Projections of future GrIS surface melt are scarce, as most global climate models do not feature a (realistic) melt calculation (Cullather et al., 2014; Lenaerts et al., 2019; Vizcaino et al., 2014).

The classical approach to use climate simulations for estimates of ice sheet surface melt is through positive-degree-day schemes (Braithwaite, 1995; Wake & Marshall, 2015). While these schemes are often used for computational efficiency, their performance is poor when surface melt/ablation is high (Bauer & Ganopolski, 2017), which we can expect for most global warming scenarios. More advanced, state-of-the-art 21st century projections of GrIS surface melt come from RCMs (Fettweis et al., 2013; Mottram et al., 2017; van Angelen et al., 2013). These models are run at high resolution and with a surface energy balance-based calculation of melt. However, they are computationally expensive and require external forcing from a global climate model (Noël et al., 2020).

Here, we investigate the performance of artificial neural networks (ANNs) in translating climate projections from the global models participating in the Climate Model Intercomparison Project Phase 6 (CMIP6; Eyring et al., 2016) into GrIS surface melt projections. ANNs are computationally efficient and able to learn complex nonlinear relationships. We explore whether ANNs can learn the relationship between quantities





simulated by a global climate model and simulated melt through training with output from the Community Earth System Model version 2.1 (CESM2; Danabasoglu et al., 2020). CESM2 features an advanced, interactive calculation of the GrIS surface melt based on downscaling of the surface energy balance (Sellevold et al., 2019; van Kampenhout et al., 2017, 2020). This study aims to take the first steps toward further use of global climate models and ANNs to project GrIS surface melt.

2. Methods

2.1. Climate Data for Training

To train the neural networks, we use the model output from CESM2 (Danabasoglu et al., 2020), an advanced Earth system model with components for atmosphere, ocean, sea-ice, land, and (optionally) land ice. It is run with a nominal l° horizontal resolution. Further, it features a surface melt calculation in the land component with advanced snow physics (Flanner & Zender, 2006; van Kampenhout et al., 2017) and elevation classes downscaling (Sellevold et al., 2019). This calculation is evaluated in van Kampenhout et al. (2020) and applied to simulate future evolution of GrIS SMB in Sellevold and Vizcaíno (2020) and GrIS dynamical evolution in and Muntjewerf, Petrini, et al. (2020).

We use annual CESM2 melt rates from 10 historical and 19 scenario simulations (Table S2 and Figure S1). These CESM2 simulation years were randomly divided into two parts: the first to train the neural networks (90%), and the second for cross-validation (that is, to tune the neural networks, 10%).

2.2. Artificial Neural Network

We designed the ANNs to predict annual melt by independently using summer (JJA) averages of five atmospheric variables: near-surface temperature (T_{2m}) , 500 hPa geopotential heights (Z_{500}), cloud cover (CC), incoming radiation (RAD_{in}; the sum of incoming shortwave and longwave radiation), and snowfall (SNOW). These variables were chosen as they have been identified to be connected to melt increase (Hanna et al., 2016; Hofer et al., 2017; Tedesco & Fettweis, 2020a). After preliminary testing, we decided to train an ANN per variable versus a combination of variables to avoid the network focusing on one variable more than others, as individual variables may be biased in CESM2 or other climate models. Further, using individual variables for each ANN allows for uncertainty quantification based on variable selection.

Before feeding the climate data to the ANNs, we apply a preprocessing step. The data is scaled (Table S1) to the range 0.2–0.8 to make the network find the optimal combination of weights more efficiently. Further, the data is transformed from two-dimensional maps to a one-dimensional vector. The resulting vector is of size n = number of latitude points × number of longitude points = 192 × 288 = 55,296. The maps are global rather than cropped to a user-defined region around Greenland, as we want the neural networks to determine their region(s) of interest.

The ANNs consist of an input layer, a hidden layer, and an output layer (schematic of architecture in Figure S2). The input layer is the first layer of the ANN and takes in the vectorized data. The input is then passed to each of the units in the hidden layer. The hidden layer consists of four units. This number of units was chosen after testing with different initialization methods and hyperparameters, where the number of units activated by the networks never exceeded four. Each unit (*i*) holds its unique weight vector W_i of same size as \vec{X} . These weights are randomly initialized (Glorot & Bengio, 2010). For each unit, we calculate a scalar F_i:

$$F_i(\overrightarrow{W_i}, \overrightarrow{X}) = B_i + \overrightarrow{W_i^T} \cdot \overrightarrow{X},\tag{1}$$

where B_i is the bias. We then apply an activation function, which transforms F_i to zero if $F_i < 0$, and does nothing if $F_i \ge 0$ (also known as rectified linear unit activation; Xu et al., 2015).

When optimizing these weights, we use simple ridge regression (also known as Tikhonov regularization with units regularization matrix or L2 regularization). Ridge regression applies a penalty (λ ; Table S1) for large weights during optimization to ensure that weights do not grow too large. This is a common technique to avoid overfitting and to derive physically coherent patterns (Barnes et al., 2019). The weights of each



unit can be restructured into the gridded format of the training data. Weights restructured in this way are referred to as feature maps.

The output layer gives the predicted melt rates as the annual integrated GrIS melt. It predicts the melt through

$$MELT = b + \sum_{i=1}^{4} F_i \cdot w_i \tag{2}$$

where *b* is a bias term, and w_i are the weights in the output layer. They are calculated as in the hidden layer, though, without the use of an activation function.

Training of the network is done through a feedforward and backpropagation algorithm. The data is sent through the network in batches of 30 samples in the feedforward. One sample is equal to one summer average of the input variable with its corresponding annual melt. The samples in each batch are selected randomly from the available training data, so the network does not learn the time order. The network generates a melt prediction, and the mean squared error between the predicted melt and the melt as simulated by CESM2 is calculated. To minimize the error, backpropagation and gradient descent are performed. Backpropagation computes the gradient of the loss function (mean squared error) with respect to the weights efficiently. We use the adaptive moment estimation (ADAM) gradient descent algorithm to optimize the ANNs' weights to minimize the loss function. ADAM is a stochastic algorithm with exponential decay of the learning rate (Kingma & Ba, 2014). Here, we use learning rates in the range of $10^{-3} - 10^{-4}$ (Table S1). We use 0.9 for the first moment exponential decay rate and 0.999 for the second raw moment exponential decay rate.

Once all available batches of samples are through the network, an epoch has ended. Before starting a new epoch, the samples are randomly shuffled and divided into new batches. We let the ANNs train for 20,000 epochs and save the ANNs' state where the melt predictions are the closest to the explicitly calculated melt for the cross-validation data. The number of epochs needed was in the range of 630–19,757 (Table S1), depending on the variable.

2.3. Climate Data for Melt Projection

We use the output from all global climate models participating in the CMIP6 to project GrIS surface melt for the historical period, and four Shared Socioeconomic Pathways (SSPs; O'Neill et al., 2016) (Table S2). In total, we use 345, 123, 147, 147, and 129 simulations from 46, 30, 32, 28, and 32 climate models for the historical period, SSP1-2.6, SSP2-4.5, SSP3-7.0, and SSP5-8.5, respectively.

We bilinearly interpolate the global climate model output to the CESM2 ~1° horizontal grid to be compatible with the ANNs' input layers. Further, we use climate anomalies with respect to the 1979–1998 mean climatology, which we add to the CESM2 1979–1998 ensemble-mean climatology. This avoids potential artifacts in the projections due to a different baseline climatology than used to train the ANNs. Therefore, we only assess the melt increase due to a change in the climate forcing. We do not discard any simulations if some variables are not available.

3. Results

3.1. Analysis of the Neural Networks

To demonstrate our method, we use the ANNs to predict melt from five atmospheric variables obtained from an independent CESM2 historical/SSP5-8.5 simulation not used for training or cross-validation and compare to the explicit melt calculation in this simulation. Figure 1 shows the feature maps corresponding to each activated function (the weights learned by the ANNs in the hidden layer; W_i). Here, we only show weights in the Greenland region, as weights from the other areas are much smaller (Figure S3).

All the input variables have high predictive skill when compared with the explicit melt calculation in CESM2 (see annotations in Figure 1). The correlation coefficients show that they can both capture the melt





Figure 1. Comparison of artificial neural network (ANN) melt projection (blue line) from five (summer-mean, JJA) climate variables with melt explicitly simulated (black line) for the same independent historical/SSP5-8.5 CESM2 climate simulation (Noël et al., 2020): (a) from near-surface temperature, (b) from 500 hPa geopotential heights, (c) from cloud cover, (d) from incoming radiation, and (e) from snowfall, all in Gt yr⁻¹. The orange, green, and red lines correspond to the melt contribution from the feature maps with matching frame color to the right of the time series. The sum of the colored lines equals the total ANN predicted melt (blue line). The annotated *r* shows the correlation coefficient between the explicitly calculated CESM2 melt and the ANN predictions, with the detrended (by subtracting 20 years running mean) correlation coefficient in parentheses. The root mean squared error (RMSE) for the entire time series and bias in the last 20 years of the ANN predictions are shown as a percentage of the CESM2 calculated melt in the last 20 years.

increase and the interannual variability. Further, predictions with T_{2m} and Z_{500} overestimate melt, while predictions with CC, RAD_{*in*} and SNOW underestimate melt.

An advantage of the ANNs is their ability to represent nonlinearities in the melt evolution, even if they are not present in the input data. This is achieved by applying activation functions to the hidden layer units. For instance, the function for T_{2m} represented by the green line (and corresponding to the map with the green frame) activates around 2040 and sharply increases around the years 2060–2065 (Figure 1a). 2060–2065 corresponds to the timing of acceleration of simulated melt (year 2071, according to a breakpoint analysis; Muntjewerf, Sellevold, et al., 2020).

For T_{2m} , the highest weights in the first map (orange frame) are over the GrIS, particularly at the southern dome. Also, temperatures over the adjacent ocean are weighted, indicating a connection between warmer oceans near Greenland and increased melt. All these weights are positive, indicating that rising temperatures and increased melt are connected. The feature map representing the year 2060 acceleration (green frame) weighs the interior and over the oceans more strongly. This is due to the interior of the ice sheet warming more than the margins during summer with global warming (Fettweis et al., 2013). As the ice surface cannot exceed the melting point, the warming of the atmosphere directly above is to some extent limited (Vizcaino et al., 2014). Also, Arctic Ocean temperatures close to the northeastern part are positively weighted in connection with major summer sea-ice reductions after 2060.

For Z_{500} (green line), CC (red line), and RAD_{*in*} (green line), the activation of an additional feature takes place in between 1990 and 2000 (Figures 1b–1d). Further, these activated features, or melt contributions, increase around the year 2020. These two timings correspond well to the melt increases simulated around these periods. The 2060–2065 melt increase coincides with the activation and increase of another unit for Z_{500} (red line).

The connection between Z_{500} and melt is well established (Box et al., 2012; Delhasse et al., 2018; Hanna et al., 2016; Sellevold & Vizcaíno, 2020). Increased Z_{500} promotes more heat advection, clear skies (Hofer et al., 2017) and blocking. Also, negative Z_{500} anomalies, indicative of more cyclones, are linked to melting by increasing rainfall over the GrIS (Oltmanns et al., 2019). The melt prediction strongly weights Z_{500} over the GrIS (Figure 1b, orange frame). Z_{500} is also negatively weighted over the Labrador Sea. For the post-2000 activation (Figure 1b, green frame) similar, though less strong, patterns appear. A third activation happens around the year 2065, with the associated patterns in Figure 1b, red frame. In contrast to the previous feature maps, the center of the positive weights over the GrIS moves from central Greenland to the eastern coast and a weaker center in the northwest. Further, the negative weights over the Labrador Sea intensify.

The CC maps that are active from the beginning of the ANN prediction are very similar (Figure 1c, orange and green frames). They predict more melt if cloud cover decreases around the margins or more melt if cloud cover increases in the interior. This is likely due to the competing effects of increased incoming short-wave and decreased incoming longwave radiation with decreasing cloud cover, and vice versa (Izeboud et al., 2020; Wang et al., 2019). For high albedo surfaces, such as the snow-covered interior, reduced cloud cover increases incoming shortwave radiation for a larger area. However, since a high amount of this radiation is reflected, this has minimal impact on the surface energy balance. If cloud cover increases, incoming longwave radiation increases, resulting in more energy at the surface. For a low albedo surface, such as the bare ice exposed margins, the effect is opposite, and thus decreasing the cloud cover increases the energy at the surface more than if cloud cover increases. From around the year 2000, a third feature map is activated (Figure 1c, red line and frame), which is also similar to the previously activated feature maps.

The two feature maps for RAD_{in} are very similar (Figure 1d). They show that increasing RAD_{in} over the GrIS surface increases the projected melt. Interestingly, the weights show that increased RAD_{in} in the south— southeast is the strongest indicator for melt increase. A plausible explanation for the similar feature maps is that the evolution of incoming radiation is rather linear, so it cannot account for the nonlinear increase in the surface melt. So another feature needs to activate to account for the nonlinear increase in the melt, likely due to lowering the GrIS' surface albedo with elevated global warming.

The melt prediction based on SNOW (Figure 1e) shows that more melt is predicted with reduced SNOW at the margins and increased snow in the interior. Less summer snowfall at the margins is linked to higher

melt rates (Noël et al., 2015; Tedesco & Fettweis, 2020b), by not temporarily increasing the albedo and interrupting bare ice exposure. For the interior, this pattern is likely aliasing of the evolution due to a warming world: increasing the global temperatures leads to more snowfall in the interior (Mottram et al., 2017).

One interpretation of the activation of additional features can be that the melt accelerates when they are activated (e.g., T_{2m} and Z_{500} , Figures 1a and 1b). However, this also depends on the input variable. If the input variable follows a similar evolution as melt, the surface melt is predicted without activation of additional features (e.g., SNOW). So another interpretation of the activation features can be that they represent feedback mechanisms, such as the melt-albedo feedback (Box et al., 2012), that accelerates melt as ablation areas expand (Muntjewerf, Sellevold, et al., 2020; Sellevold & Vizcaíno, 2020).

3.2. Comparison With Regional Climate Modeling

We compare the GrIS melt from our ANNs to melt calculated by the Modele Atmospherique Regional (MAR, Hanna et al., 2020), one of the most advanced RCM for studying GrIS surface processes, which has been extensively evaluated for studies of the GrIS (Fettweis et al., 2017). The aim is to test the quality of the melt estimate from ANNs. MAR dynamically downscaled a historical and SSP5-8.5 (plus SSP1-2.6 for CNRM-CM-6-1) climate simulation from five global climate models (Figure 2). By comparing the melt predicted by each of the ANNs and the simulated melt from MAR, we find that the ANNs using Z_{500} , CC, and RAD_{in} did not perform well (Figure S4). Likely, the reasons are that these variables are known to be uncertain in projections and dependent on the model physics. If an atmospheric variable is simulated very differently in the RCM versus the global climate model used for forcing, the relationship between this variable and melt in the ANN may not correspond to the relationship in the RCM. On the other hand, T_{2m} and SNOW are likely more influenced by large-scale dynamics, which will not differ much in the RCM compared to the global model used as lateral forcing (Ettema et al., 2010). Therefore, further in this comparison, we use the mean of the melt produced by the ANNs trained with T_{2m} and SNOW.

The ANNs' integrated melt correlates very well to the melt simulated by MAR for all the different climate models (Figures 2b–2e). The mean of all the models (Figure 2f) shows a correlation as strong as the correlation using CESM2. Except for CNRM-ESM2-1, the interannual melt variability simulated by MAR is also well captured by the ANNs. The RMSE for individual SSP5-8.5 simulations is in the range of 8.4–16.1% (60.0% of the SSP1-2.6 simulation). The primary cause of this error is the ANNs producing lower interannual melt variability magnitude due to the ANNs being a mean of the ANN for T_{2m} and SNOW. The RMSE for the mean of all models (both for MAR and the ANNs) is lower than for individual simulations.

For CESM2, the ANNs generate an end-of-the-century mean melt that is, 15.6% lower than what is simulated by MAR. Keeping in mind that the ANNs captured this metric well for CESM2, this bias's leading cause is the difference in melt simulated by MAR and CESM2 for the same (global) climate simulation. The models with lower climate sensitivity than CESM2 (CNRM-CM6-1, CNRM-ESM2-1, and MRI-ESM2-0), the ANNs have a positive biased end-of-the-century melt projection compared to MAR. The model with higher climate sensitivity than CESM2 (UKESM1-0-LL) has a negative bias compared to MAR, although smaller (-1.7%) than the CESM2 bias. The percentage bias for the SSP1-2.6 is much larger than for the SSP5-8.5, with the ANNs projecting a 71.9% higher melt increase than MAR for the SSP1-2.6 simulation. When comparing the mean of all the SSP5-8.5 simulations with MAR and the ANNs, the bias is small (1.6%; Figure 2f), and the MAR simulated melt is within the ANNs' model spread.

3.3. Surface Melt Projections

The global mean T_{2m} evolution for the CMIP6 model ensemble (Table S2) using 1979–1998 as a baseline is shown in Figure 3a. The ensemble mean shows increasing temperatures since the 1960s (Figure 3a). All the scenarios show significant (p < 0.01) warming in the last 20 years of the simulation (Table 1), except for SSP1-2.6.

Using the T_{2m} and SNOW ANNs, the global warming simulation can be translated into GrIS surface melt (Figure 3b and Table 1). For the historical period, surface melt starts to increase around 1975. The ANNs project the melt to increase in the range of 79% (SSP1-2.6) to 264% (SSP5-8.5) (Table 1) with respect to the





Figure 2. 1950–2100 time series of Greenland ice sheet (GrIS) integrated surface melt change (Gt yr⁻¹), with respect to 1979–1998, in Modele Atmospherique Regional (MAR) and the artificial neural networks (ANNs) (mean of T_{2m} and SNOW) for an individual climate simulation from (a) CNRM-CM6-1, (b) CNRM-ESM2-1, (c) UKESM1-0-LL, (d) MRI-ESM2-0, and (e) CESM2. Panel (f) shows the mean of (a–e). Black, blue, and red dotted (solid) lines correspond to the historical, SSP1-2.6, and SSP5-8.5, respectively, as projected by the ANNs (MAR). The annotated *r* shows the correlation coefficient between the explicitly calculated MAR melt and the ANN predictions, with the detrended (by subtracting 20 years running mean) correlation coefficient in parenthesis. The root mean squared error (RMSE) for the entire time series and bias in the last 20 years of the ANN predictions are shown as a percentage of the MAR simulated melt change in the last 20 years.



Figure 3. Climate Model Intercomparison Project Phase 6 (CMIP6) mean global mean temperature and Greenland ice sheet (GrIS) surface melt change in the historical period and the four Shared Socioeconomic Pathways (SSPs) for (a) T_{2m} change (*k*) and (b) GrIS surface melt change (Gt yr⁻¹) with respect to the 1979–1998 mean. (c) Source of melt change uncertainty in the projections. For panels (a and b), black lines show the historical period, blue lines show SSP1-2.6, orange lines show SSP2-4.5, green lines show SSP3-7.0, and red lines show SSP5-8.5. The shading represents the model spread at ±0.5 standard deviation (instead of 1 standard deviation to avoid heavy overlapping). In panel (c), the purple line corresponds to the variable spread, olive line to internal climate spread, cyan line to model spread, and pink line to scenario spread. The spread is given by one standard deviation. Further details on uncertainty quantification are given in Text S1.

Table 1	
---------	--

Surface Melt Projections From Community Earth System Model 2.1 (CESM2) (Explicitly Simulated) and the Artificial Neural Networks (ANNs) Using the Mean of T_{2m} and SNOW

Scenario	Global mean T_{2m}	CESM2	ANNs evaluated on CMIP6
Historical (mean)	16.0 ± 0.1	447 ± 90	521 ± 63
SSP1-2.6 (change)	1.6 ± 0.5	413 ± 95	414 ± 276
SSP2-4.5 (change)	2.5 ± 0.6	619 ± 140	724 ± 371
SSP3-7.0 (change)	3.5 ± 0.8	$1,040 \pm 170$	$1,031 \pm 436$
SSP5-8.5 (change)	4.4 ± 1.0	$1,834 \pm 152$	$1,378 \pm 555$

Note. "Historical" represents the 1979–1998 mean. For SSP's, 2081–2100 anomalies with respect to Historical are given. The \pm denotes one standard deviation model spread for the last 20 years. The column ANNs gives the results for the full range of CMIP6 models. With the exception of the global mean T_{2m} (K), all numbers are of GrIS surface melt (Gt yr⁻¹).

1979–1998 mean. Surface melt diverges between the years 2030 and 2050 for the different scenarios. By the end of the century, all SSP scenarios except for SSP1-2.6 show a significant positive trend, meaning they will likely continue to increase beyond the 21st century. SSP1-2.6 shows a nonsignificant (p = 0.02) decrease. Generally, the ANNs project a higher surface melt increase for models with a higher increase in global mean temperature (Figure S5).

The projected surface melt increase by the ANNs agrees well with the explicitly simulated melt increase by CESM2 (Table 1) for all scenarios except for SSP5-8.5. This difference is likely due to the large range of climate sensitivities of the CMIP6 models (Zelinka et al., 2020), augmented by the different regional simulations for a given global mean temperature increase. Also, the ANNs project higher uncertainties for each scenario than CESM2.

The primary source of projection uncertainty is the climate model spread (Figure 3c), defined as the standard deviation of the climate model mean melt calculated by the ANNs (Text S1). The climate models' different climate sensitivity (Figure S5) is the primary cause of the large spread. The uncertainty related to scenario spread (defined as the standard deviation of the scenario's mean melt) is smaller than the model spread throughout the 21st century but becomes similar in the last years. The uncertainties associated with internal climate variability and variable selection (T_{2m} and SNOW) are smaller than the sources mentioned above. While the spread due to internal climate variability remains relatively constant, the spread due to variable selection increases from 2040.

4. Discussion and Conclusions

In this study, we have trained ANNs with data from a global climate model, which includes an explicit simulation of ice sheets SMB (CESM2) to simulate 21st century GrIS surface melt from the atmospheric output of the full CMIP6 suite of climate models. The ANNs show good performance in predicting melt compared to the explicit modeling in an independent CESM2 simulation and simulations with the RCM MAR for a subset of CMIP6 models. By using the full suite of CMIP6 models, we can reassess the uncertainty of future melt estimates. Our results provide a larger uncertainty (+87 Gt yr⁻¹) for the melt corresponding to the SSP5-8.5 scenario than a recent estimate from RCMs (Hanna et al., 2020), likely due to the inclusion of additional climate models and simulations per scenario. We find that the model spread is the primary source of uncertainty in the projections. By the end of the century, uncertainty due to scenario spread becomes an equal contributor to projection uncertainty.

The ANNs project GrIS surface melt increase by 414 ± 276 , 724 ± 371 , $1,031 \pm 436$, and $1,378 \pm 555$ Gt yr⁻¹ for SSP1-2.6, SSP2-4.5, SSP3-7.0, and SSP5-8.5, respectively. These projected increases here are higher than previous estimates, which give melt increases of 648 Gt yr⁻¹ (van Angelen et al., 2013) for Representative Concentration Pathway (RCP) 4.5 and 634 Gt yr⁻¹ for RCP8.5 (Vizcaino et al., 2014). The higher melt



Acknowledgments

R. Sellevold acknowledges support from the Nederlandse Organisatie voor Wetenschappelijk Onderzoek (grant no. ALWOP.2015.096), and M. Vizcaino from the European Research Council (grant no. ERC-StG-678145-CoupledIceClim). The independent CESM climate simulations were performed by L. van Kampenhout on SURFsara HPC systems with support from NWO Exacte Wetenschappen. RACMO2.3p2 data were provided by Brice Noel (Noël et al., 2020). We acknowledge the World Climate Research Program, which, through its Working Group on Coupled Modeling, coordinated and promoted CMIP6. We thank the climate modeling groups for producing and making available their model output, the Earth System Grid Federation (ESGF) for archiving the data and providing access, and the multiple funding agencies who support CMIP6 and ESGF (data from CMIP6 used in this study; Bader et al., 2019a, 2019b, 2020; (EC-Earth), 2019a, 2019b, 2019c, 2019d, 2019e, 2019f, 2019g, 2019h, 2019i, 2019j; Bentsen et al., 2019a, 2019b, 2019c, 2019d, 2019e; Boucher et al., 2018, 2019a, 2019b, 2019c, 2019d; Cao, 2019a, 2019b, 2019c; Cao & Wang, 2019; Chai, 2020; Danabasoglu, 2019b, 2019c, 2019d, 2019e, 2019f, 2019a, 2019h, 2019i, 2019j, 2019k, 2019l, 2019g; Dix et al., 2019a, 2019b, 2019c, 2019d, 2019e; Good, 2020a, 2019, 2020b; Good et al., 2019a, 2019b, 2019c, 2019d; Guo et al., 2018a, 2018b, 2018c; Hajima et al., 2019; John et al., 2018a, 2018b, 2018c, 2018d; Jungclaus et al., 2019; Krasting et al., 2018; Lee & Liang, 2020; Li, 2019a, 2019b, 2019c, 2019d, 2019e; NASA/GISS, 2018, 2020a, 2020b, 2020c, 2020d, 2019a, 2019b; Neubauer et al., 2019; Park & Shin, 2019; Ridley et al., 2019a, 2019b; Rong, 2019; Schupfner et al., 2019a, 2019b, 2019c, 2019d; Seferian, 2018; Seland et al., 2019a, 2019b, 2019c, 2019d, 2019e; Semmler et al., 2018a, 2018b, 2018c, 2019a, 2019b; Shiogama et al., 2019a, 2019b, 2019c, 2019d; Song et al., 2019a, 2019b, 2019c, 2019d; , Swart et al., 2019a, 2019b, 2019c, 2019d, 2019e, 2019f, 2019g, 2019h, 2019i, 2019j; Tachiiri et al., 2019a, 2019b, 2019c, 2019d; Tang et al., 2019; Tatebe & Watanabe, 2018; Voldoire, 2018, 2019a, 2019b, 2019c, 2019d, 2019e, 2019f, 2019g, 2019h, 2019i, 2019j, 2019k, 2020a, 2020b; Volodin et al., 2019a, 2019b, 2019c, 2019d, 2019e, 2019f, 2019g, 2019h, 2019i, 2019j; Wieners et al., 2019a, 2019b, 2019c, 2019d, 2019e; Wu et al., 2018; Xin et al., 2019a, 2019b, 2019c, 2019d; Yu, 2019a, 2019b, 2019c, 2019d, 2019e; Yukimoto et al., 2019a, 2019b, 2019c, 2019d, 2019e; Zhang et al., 2018; Ziehn et al., 2019a, 2019b, 2019c, 2019d, 2019e).

increases projected here are likely connected with the higher climate sensitivity of the CMIP6 models when compared to the CMIP5 models (Meehl et al., 2020; Zelinka et al., 2020).

We opted for a simple ANN structure in this study, as it allows us to explore the feature maps and activations easily. More advanced ANN structures, like multichannel deep convolutional neural networks (Barros et al., 2014), which can draw connections between variables, could be applied in a follow-up for this study for comparison of projections.

Data Availability Statement

Projected melt from the ANNs can be obtained from https://doi.org/10.5281/zenodo.4424895, and the code used for this manuscript, as well as the models, are available from https://doi.org/10.5281/zenodo.5153908.

References

Bader, D. C., Leung, R., Taylor, M., & McCoy, R. B. (2019a). E3SM-Project E3SM1.0 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4497

- Bader, D. C., Leung, R., Taylor, M., & McCoy, R. B. (2019b). E3SM-Project E3SM1.1 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.11485
- Bader, D. C., Leung, R., Taylor, M., & McCoy, R. B. (2020). E3SM-Project E3SM1.1ECA model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.11486

Bamber, J. L., Westaway, R. M., Marzeion, B., & Wouters, B. (2018). The land ice contribution to sea level during the satellite era. Environmental Research Letters, 13(6), 063008. https://doi.org/10.1088/1748-9326/aac2f0

Barnes, E. A., Hurrell, J. W., Ebert-Uphoff, I., Anderson, C., & Anderson, D. (2019). Viewing forced climate patterns through an AI lens. Geophysical Research Letters, 46, 13389–13398. https://doi.org/10.1029/2019GL084944

Barros, P., Magg, S., Weber, C., & Wermter, S. (2014). A multichannel convolutional neural network for hand posture recognition. In S. Wermter, C. Weber, W. Duch, T. Honkela, P. Koprinkova-Hristova, S. Magg, et al. (Eds.), Artificial neural networks and machine learning—ICANN 2014 (pp. 403–410). Springer International Publishing. https://doi.org/10.1007/978-3-319-11179-7_51

Bauer, E., & Ganopolski, A. (2017). Comparison of surface mass balance of ice sheets simulated by positive-degree-day method and energy balance approach. *Climate of the Past*, 13(7), 819–832. https://doi.org/10.5194/cp-13-819-2017

Bentsen, M., Oliviè, D. J. L., Seland, Ø., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019a). NCC NorESM2-MM model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8040

Bentsen, M., Oliviè, D. J. L., Seland, Ø., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019b). NCC NorESM2-MM model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8250

Bentsen, M., Oliviè, D. J. L., Seland, Ø., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019c). NCC NorESM2-MM model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8255 Bentsen, M., Oliviè, D. J. L., Seland, Ø., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019d). NCC NorESM2-MM model output

prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8270 Bentsen, M., Oliviè, D. J. L., Seland, Ø., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019e). NCC NorESM2-MM model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8221

Boucher, O., Denvil, S., Caubel, A., & Foujols, M. A. (2018). IPSL IPSL-CM6A-LR model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5195

Boucher, O., Denvil, S., Caubel, A., & Foujols, M. A. (2019a). IPSL IPSL-CM6A-LR model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5262

- Boucher, O., Denvil, S., Caubel, A., & Foujols, M. A. (2019b). *IPSL IPSL-CM6A-LR model output prepared for CMIP6 ScenarioMIP ssp245* (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5264
- Boucher, O., Denvil, S., Caubel, A., & Foujols, M. A. (2019c). *IPSL IPSL-CM6A-LR model output prepared for CMIP6 ScenarioMIP ssp370* (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5265
- Boucher, O., Denvil, S., Caubel, A., & Foujols, M. A. (2019d). *IPSL IPSL-CM6A-LR model output prepared for CMIP6 ScenarioMIP ssp585* (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5271

Box, J. E., Fettweis, X., Stroeve, J. C., Tedesco, M., Hall, D. K., & Steffen, K. (2012). Greenland ice sheet albedo feedback: Thermodynamics and atmospheric drivers. *The Cryosphere*, 6(4), 821–839. https://doi.org/10.5194/tc-6-821-2012

Braithwaite, R. J. (1995). Positive degree-day factors for ablation on the Greenland ice sheet studied by energy-balance modelling. *Journal* of Glaciology, 41(137), 153–160. https://doi.org/10.3189/s0022143000017846

Cao, J. (2019a). NUIST NESMv3 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8780

Cao, J. (2019b). NUIST NESMv3 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8781

Cao, J. (2019c). NUIST NESMv3 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8790

Cao, J., & Wang, B. (2019). NUIST NESMv3 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8769

Chai, Z. (2020). CAS CAS-ESM1.0 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3353

Cullather, R. I., Nowicki, S. M. J., Zhao, B., & Suarez, M. J. (2014). Evaluation of the surface representation of the Greenland ice sheet in a General Circulation Model. *Journal of Climate*, *27*(13), 4835–4856. https://doi.org/10.1175/JCLI-D-13-00635.1



- Danabasoglu, G. (2019a). NCAR CESM2-FV2 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.11297
- Danabasoglu, G. (2019b). NCAR CESM2 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7627
- Danabasoglu, G. (2019c). NCAR CESM2 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7746
- Danabasoglu, G. (2019d). NCAR CESM2 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7748
- Danabasoglu, G. (2019e). NCAR CESM2 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7753
- Danabasoglu, G. (2019f). NCAR CESM2 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7768
- Danabasoglu, G. (2019g). NCAR CESM2-WACCM-FV2 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.11298
- Danabasoglu, G. (2019h). NCAR CESM2-WACCM model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.10071
- Danabasoglu, G. (2019i). NCAR CESM2-WACCM model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.10100
- Danabasoglu, G. (2019j). NCAR CESM2-WACCM model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.10101
- Danabasoglu, G. (2019k). NCAR CESM2-WACCM model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.10102
- Danabasoglu, G. (2019l). NCAR CESM2-WACCM model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.10115
- Danabasoglu, G., Lamarque, J.-F., Bacmeister, J., Bailey, D. A., DuVivier, A. K., Edwards, J., et al. (2020). The Community Earth System Model Version 2 (CESM2). Journal of Advances in Modeling Earth Systems, 12, e2019MS001916. https://doi.org/10.1029/2019MS001916
- Delhasse, A., Fettweis, X., Kittel, C., Amory, C., & Agosta, C. (2018). Brief communication: Impact of the recent atmospheric circulation change in summer on the future surface mass balance of the Greenland ice sheet. *The Cryosphere*, *12*(11), 3409–3418. https://doi. org/10.5194/tc-12-3409-2018
- Dix, M., Bi, D., Dobrohotoff, P., Fiedler, R., Harman, I., Law, R., & Yang, R. (2019a). CSIRO-ARCCSS ACCESS-CM2 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4271
- Dix, M., Bi, D., Dobrohotoff, P., Fiedler, R., Harman, I., Law, R., & Yang, R. (2019b). CSIRO-ARCCSS ACCESS-CM2 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4319
- Dix, M., Bi, D., Dobrohotoff, P., Fiedler, R., Harman, I., Law, R., & Yang, R. (2019c). CSIRO-ARCCSS ACCESS-CM2 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4321
- Dix, M., Bi, D., Dobrohotoff, P., Fiedler, R., Harman, I., Law, R., & Yang, R. (2019d). CSIRO-ARCCSS ACCESS-CM2 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4323
- Dix, M., Bi, D., Dobrohotoff, P., Fiedler, R., Harman, I., Law, R., & Yang, R. (2019e). CSIRO-ARCCSS ACCESS-CM2 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4332
- EC-Earth, E.-E.C. (2019a). EC-Earth-Consortium EC-Earth3 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4700
- EC-Earth, E.-E.C. (2019b). EC-Earth-Consortium EC-Earth3 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4874
- EC-Earth, E.-E.C. (2019c). EC-Earth-Consortium EC-Earth3 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4880
- EC-Earth, E.-E.C. (2019d). EC-Earth-Consortium EC-Earth3 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation.
- EC-Earth, E.-E.C. (2019e). EC-Earth-Consortium EC-Earth3 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4912
- EC-Earth, E.-E.C. (2019f). EC-Earth-Consortium EC-Earth3-Veg model output prepared for CMIP6 CMIP historical (Version 20200512). Earth system grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4706
- EC-Earth, E.-E.C. (2019g). EC-Earth-Consortium EC-Earth3-Veg model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4876
- EC-Earth, E.-E.C. (2019h). EC-Earth-Consortium EC-Earth3-Veg model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4882
- EC-Earth, E.-E.C. (2019i). EC-Earth-Consortium EC-Earth3-Veg model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4886
- EC-Earth, E.-E.C. (2019j). EC-Earth-Consortium EC-Earth3-Veg model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4914
- Enderlin, E. M., Howat, I. M., Jeong, S., Noh, M.-J., van Angelen, J. H., & van den Broeke, M. R. (2014). An improved mass budget for the Greenland ice sheet. *Geophysical Research Letters*, *41*, 866–872. https://doi.org/10.1002/2013GL059010
- Ettema, J., van den Broeke, M. R., van Meijgaard, E., & van de Berg, W. J. (2010). Climate of the Greenland ice sheet using a high-resolution climate model—Part 2: Near-surface climate and energy balance. *The Cryosphere*, 4(4), 529–544. https://doi.org/10.5194/tc-4-529-2010
- Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E. (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental design and organization. *Geoscientific Model Development*, 9(5), 1937–1958. https:// doi.org/10.5194/gmd-9-1937-2016
- Fettweis, X., Box, J. E., Agosta, C., Amory, C., Kittel, C., Lang, C., et al. (2017). Reconstructions of the 1900–2015 Greenland ice sheet surface mass balance using the regional climate MAR model. *The Cryosphere*, *11*(2), 1015–1033. https://doi.org/10.5194/tc-11-1015-2017
- Fettweis, X., Franco, B., Tedesco, M., van Angelen, J. H., Lenaerts, J. T. M., van den Broeke, M. R., & Gallée, H. (2013). Estimating the Greenland ice sheet surface mass balance contribution to future sea level rise using the regional atmospheric climate model MAR. *The Cryosphere*, 7(2), 469–489. https://doi.org/10.5194/tc-7-469-2013



Flanner, M. G., & Zender, C. S. (2006). Linking snowpack microphysics and albedo evolution. Journal of Geophysical Research, 111, D12208. https://doi.org/10.1029/2005JD006834

- Glorot, X., & Bengio, Y. (2010). Understanding the difficulty of training deep feedforward neural networks. In Proceedings of the Thirteenth International Conference on Artificial Intelligence and Statistics (Vol. 9, pp. 249–256). PMLR.
- Good, P. (2019). MOHC HadGEM3-GC31-LL model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.10851
- Good, P. (2020a). MOHC HadGEM3-GC31-LL model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.10849
- Good, P. (2020b). MOHC HadGEM3-GC31-LL model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.10901
- Good, P., Sellar, A., Tang, Y., Rumbold, S., Ellis, R., Kelley, D., & Kuhlbrodt, T. (2019a). MOHC UKESM1.0-LL model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6333
- Good, P., Sellar, A., Tang, Y., Rumbold, S., Ellis, R., Kelley, D., & Kuhlbrodt, T. (2019b). MOHC UKESM1.0-LL model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6339
- Good, P., Sellar, A., Tang, Y., Rumbold, S., Ellis, R., Kelley, D., & Kuhlbrodt, T. (2019c). MOHC UKESM1.0-LL model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6347
- Good, P., Sellar, A., Tang, Y., Rumbold, S., Ellis, R., Kelley, D., & Kuhlbrodt, T. (2019d). MOHC UKESM1.0-LL model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6405
- Guo, H., John, J. G., Blanton, C., McHugh, C., Nikonov, S., Radhakrishnan, A., & Zhang, R. (2018a). NOAA-GFDL GFDL-CM4 model output historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8594
- Guo, H., John, J. G., Blanton, C., McHugh, C., Nikonov, S., Radhakrishnan, A., & Zhang, R. (2018b). NOAA-GFDL GFDL-CM4 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.9263
- Guo, H., John, J. G., Blanton, C., McHugh, C., Nikonov, S., Radhakrishnan, A., & Zhang, R. (2018c). NOAA-GFDL GFDL-CM4 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.9268
- Hajima, T., Abe, M., Arakawa, O., Suzuki, T., Komuro, Y., Ogura, T., & Tachiiri, K. (2019). MIROC MIROC-ES2L model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5602
- Hanna, E., Cappelen, J., Fettweis, X., Mernild, S. H., Mote, T. L., Mottram, R., & Hall, R. J. (2020). Greenland surface air temperature changes from 1981 to 2019 and implications for ice-sheet melt and mass-balance change. *International Journal of Climatology*. 41(S1), E1336–E1352. https://doi.org/10.1002/joc.6771
- Hanna, E., Cropper, T. E., Hall, R. J., & Cappelen, J. (2016). Greenland Blocking Index 1851–2015: A regional climate change signal. International Journal of Climatology, 36(15), 4847–4861. https://doi.org/10.1002/joc.4673
- Hofer, S., Tedstone, A. J., Fettweis, X., & Bamber, J. L. (2017). Decreasing cloud cover drives the recent mass loss on the Greenland ice sheet. *Science Advances*, 3(6), e1700584. https://doi.org/10.1126/sciadv.1700584
- Izeboud, M., Lhermitte, S., van Tricht, K., Lenaerts, J. T. M., van Lipzig, N. P. M., & Wever, N. (2020). The spatiotemporal variability of cloud radiative effects on the Greenland ice sheet surface mass balance. *Geophysical Research Letters*, 47, e2020GL087315. https://doi. org/10.1029/2020GL087315
- John, J. G., Blanton, C., McHugh, C., Radhakrishnan, A., Rand, K., Vahlenkamp, H., & Zeng, Y. (2018a). NOAA-GFDL GFDL-ESM4 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.8684
- John, J. G., Blanton, C., McHugh, C., Radhakrishnan, A., Rand, K., Vahlenkamp, H., & Zeng, Y. (2018b). NOAA-GFDL GFDL-ESM4 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.8686
- John, J. G., Blanton, C., McHugh, C., Radhakrishnan, A., Rand, K., Vahlenkamp, H., & Zeng, Y. (2018c). NOAA-GFDL GFDL-ESM4 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.8691
- John, J. G., Blanton, C., McHugh, C., Radhakrishnan, A., Rand, K., Vahlenkamp, H., & Zeng, Y. (2018d). NOAA-GFDL GFDL-ESM4 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.8706
- Jungclaus, J., Bittner, M., Wieners, K.-H., Wachsmann, F., Schupfner, M., Legutke, S., & Roeckner, E. (2019). MPI-M MPI-ESM1.2-HR model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.6594
- Kingma, D. P., & Ba, J. (2014). Adam: A method for stochastic optimization. Cornell University.
- Krasting, J. P., John, J. G., Blanton, C., McHugh, C., Nikonov, S., Radhakrishnan, A., & Zhao, M. (2018). NOAA-GFDL GFDL-ESM4 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.8597
- Lee, W.-L., & Liang, H.-C. (2020). AS-RCEC TaiESM1.0 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.9755
- Lenaerts, J. T. M., Medley, B., van den Broeke, M. R., & Wouters, B. (2019). Observing and modeling ice sheet surface mass balance. *Reviews of Geophysics*, 57, 376–420. https://doi.org/10.1029/2018RG000622
- Li, L. (2019a). CAS FGOALS-g3 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3356
- Li, L. (2019b). CAS FGOALS-g3 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3465
- Li, L. (2019c). CAS FGOALS-g3 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3469
- Li, L. (2019d). CAS FGOALS-g3 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3480
- Li, L. (2019e). CAS FGOALS-g3 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3503



- Meehl, G. A., Senior, C. A., Eyring, V., Flato, G., Lamarque, J.-F., Stouffer, R. J., & Schlund, M. (2020). Context for interpreting equilibrium climate sensitivity and transient climate response from the CMIP6 Earth system models. *Science Advances*, 6(26), eaba1981. https://doi. org/10.1126/sciadv.aba1981
- Mottram, R., Boberg, F., Langen, P., Yang, S., Rodehacke, C., Christensen, J. H., & Madsen, M. S. (2017). Surface mass balance of the Greenland ice sheet in the regional climate model HIRHAM5: Present state and future prospects. *Low Temperature Sciences*, 75, 105–115. https://doi.org/10.14943/lowtemsci.75.105
- Mouginot, J., Rignot, E., Bjørk, A. A., van den Broeke, M., Millan, R., Morlighem, M., et al. (2019). Forty-six years of Greenland ice sheet mass balance from 1972 to 2018. Proceedings of the National Academy of Sciences of the United States of America, 116(19), 9239–9244. https://doi.org/10.1073/pnas.1904242116
- Muntjewerf, L., Petrini, M., Vizcaino, M., Ernani da Silva, C., Sellevold, R., Scherrenberg, M. D. W., et al. (2020). Greenland ice sheet contribution to 21st century sea level rise as simulated by the coupled CESM2.1-CISM2.1. *Geophysical Research Letters*, 47, e2019GL086836. https://doi.org/10.1029/2019GL086836
- Muntjewerf, L., Sellevold, R., Vizcaino, M., Ernani da Silva, C., Petrini, M., Thayer-Calder, K., et al. (2020). Accelerated Greenland ice sheet mass loss under high greenhouse gas forcing as simulated by the coupled CESM2.1-CISM2.1. Journal of Advances in Modeling Earth Systems, 12, e2019MS002031. https://doi.org/10.1029/2019MS002031
- NASA/GISS. (2018). NASA-GISS GISS-E2.1G model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7127
- NASA/GISS. (2019a). NASA-GISS GISS-E2-1-G-CC model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.11762
- NASA/GISS. (2019b). NASA-GISS GISS-E2.1H model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7128
- NASA/GISS. (2020a). NASA-GISS GISS-E2.1G model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7410
- NASA/GISS. (2020b). NASA-GISS GISS-E2.1G model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7415
- NASA/GISS. (2020c). NASA-GISS GISS-E2.1G model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7426
- NASA/GISS. (2020d). NASA-GISS GISS-E2.1G model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7460
- Neubauer, D., Ferrachat, S., Siegenthaler-Le Drian, C., Stoll, J., Folini, D. S., Tegen, I., & Lohmann, U. (2019). HAMMOZ-Consortium MPI-ESM1.2-HAM model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi. org/10.22033/ESGF/CMIP6.5016
- Noël, B., van De Berg, W. J., van Meijgaard, E., Kuipers Munneke, P., van De Wal, R. S., & van den Broeke, M. R. (2015). Evaluation of the updated regional climate model RACMO2.3: Summer snowfall impact on the Greenland Ice Sheet. *The Cryosphere*, 9(5), 1831–1844. https://doi.org/10.5194/tc-9-1831-2015
- Noël, B., van Kampenhout, L., van de Berg, W. J., Lenaerts, J. T. M., Wouters, B., & van den Broeke, M. R. (2020). Brief communication: CESM2 climate forcing (1950–2014) yields realistic Greenland ice sheet surface mass balance. *The Cryosphere*, 14(4), 1425–1435. https:// doi.org/10.5194/tc-14-1425-2020
- Oltmanns, M., Straneo, F., & Tedesco, M. (2019). Increased Greenland melt triggered by large-scale, year-round cyclonic moisture intrusions. The Cryosphere, 13(3), 815–825. https://doi.org/10.5194/tc-13-815-2019
- O'Neill, B. C., Tebaldi, C., van Vuuren, D. P., Eyring, V., Friedlingstein, P., Hurtt, G., et al. (2016). The scenario model intercomparison project (scenariomip) for cmip6. *Geoscientific Model Development*, 9(9), 3461–3482. https://doi.org/10.5194/gmd-9-3461-2016
- Park, S., & Shin, J. (2019). SNU SAM0-UNICON model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.7789
- Ridley, J., Menary, M., Kuhlbrodt, T., Andrews, M., & Andrews, T. (2019a). MOHC HadGEM3-GC31-LL model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6109
- Ridley, J., Menary, M., Kuhlbrodt, T., Andrews, M., & Andrews, T. (2019b). MOHC HadGEM3-GC31-MM model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6112
- Rong, X. (2019). CAMS CAMS-CSM1.0 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.9754
- Schupfner, M., Wieners, K.-H., Wachsmann, F., Steger, C., Bittner, M., Jungclaus, J., & Roeckner, E. (2019a). DKRZ MPI-ESM1.2-HR model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.4397
- Schupfner, M., Wieners, K.-H., Wachsmann, F., Steger, C., Bittner, M., Jungclaus, J., & Roeckner, E. (2019b). DKRZ MPI-ESM1.2-HR model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.4398
- Schupfner, M., Wieners, K.-H., Wachsmann, F., Steger, C., Bittner, M., Jungclaus, J., & Roeckner, E. (2019c). DKRZ MPI-ESM1.2-HR model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.4399
- Schupfner, M., Wieners, K.-H., Wachsmann, F., Steger, C., Bittner, M., Jungclaus, J., & Roeckner, E. (2019d). DKRZ MPI-ESM1.2-HR model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.4403
- Seferian, R. (2018). CNRM-CERFACS CNRM-ESM2-1 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4068
- Seland, Ø., Bentsen, M., Oliviè, D. J. L., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019a). NCC NorESM2-LM model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8036
- Seland, Ø., Bentsen, M., Oliviè, D. J. L., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019b). NCC NorESM2-LM model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8248
- Seland, Ø., Bentsen, M., Oliviè, D. J. L., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019c). NCC NorESM2-LM model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8253
- Seland, Ø., Bentsen, M., Oliviè, D. J. L., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019d). NCC NorESM2-LM model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8268



Seland, Ø., Bentsen, M., Oliviè, D. J. L., Toniazzo, T., Gjermundsen, A., Graff, L. S., & Schulz, M. (2019e). NCC NorESM2-LM model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.8319
 Sellevold, R., van Kampenhout, L., Lenaerts, J. T. M., Noël, B., Lipscomb, W. H., & Vizcaino, M. (2019). Surface mass balance downscaling through elevation classes in an Earth system model: Application to the Greenland ice sheet. The Cryosphere, 13(12), 3193–3208. https://doi.org/10.5194/tc-13-3193-2019

- Sellevold, R., & Vizcaíno, M. (2020). Global warming threshold and mechanisms for accelerated Greenland ice sheet surface mass loss. Journal of Advances in Modeling Earth Systems, 12, e2019MS002029. https://doi.org/10.1029/2019MS002029
- Semmler, T., Danilov, S., Rackow, T., Sidorenko, D., Barbi, D., Hegewald, J., & Jung, T. (2018a). AWI AWI-CM1.1MR model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.2686
- Semmler, T., Danilov, S., Rackow, T., Sidorenko, D., Barbi, D., Hegewald, J., & Jung, T. (2018b). AWI AWI-CM1.1MR model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.2796
- Semmler, T., Danilov, S., Rackow, T., Sidorenko, D., Barbi, D., Hegewald, J., & Jung, T. (2018c). AWI AWI-CM1.1MR model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.2800
- Semmler, T., Danilov, S., Rackow, T., Sidorenko, D., Barbi, D., Hegewald, J., & Jung, T. (2019a). AWI AWI-CM1.1MR model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.2803
- Semmler, T., Danilov, S., Rackow, T., Sidorenko, D., Barbi, D., Hegewald, J., & Jung, T. (2019b). AWI AWI-CM1.1MR model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.2817
- Shepherd, A., Ivins, E., Rignot, E., Smith, B., van den Broeke, M., Velicogna, I., & Team, T. I. (2020). Mass balance of the Greenland ice sheet from 1992 to 2018. *Nature*, 579(7798), 233–239. https://doi.org/10.1038/s41586-019-1855-2
- Shiogama, H., Abe, M., & Tatebe, H. (2019a). MIROC MIROC6 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5743

Shiogama, H., Abe, M., & Tatebe, H. (2019b). MIROC MIROC6 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5746

- Shiogama, H., Abe, M., & Tatebe, H. (2019c). MIROC MIROC6 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5752
- Shiogama, H., Abe, M., & Tatebe, H. (2019d). MIROC MIROC6 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5771
- Song, Z., Qiao, F., Bao, Y., Shu, Q., Song, Y., & Yang, X. (2019a). FIO-QLNM FIO-ESM2.0 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.9199
- Song, Z., Qiao, F., Bao, Y., Shu, Q., Song, Y., & Yang, X. (2019b). FIO-QLNM FIO-ESM2.0 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.9208
- Song, Z., Qiao, F., Bao, Y., Shu, Q., Song, Y., & Yang, X. (2019c). FIO-QLNM FIO-ESM2.0 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.9209
- Song, Z., Qiao, F., Bao, Y., Shu, Q., Song, Y., & Yang, X. (2019d). FIO-QLNM FIO-ESM2.0 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.9214
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019a). CCCma CanESM5-CanOE model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.10260
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019b). CCCma CanESM5-CanOE model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.10269
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019c). CCCma CanESM5-CanOE model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.10270
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019d). CCCma CanESM5-CanOE model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.10271
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019e). CCCma CanESM5-CanOE model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/ CMIP6.10276
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019f). CCCma CanESM5 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3610
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019g). CCCma CanESM5 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3683
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019h). CCCma CanESM5 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3685
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019i). CCCma CanESM5 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3690
- Swart, N. C., Cole, J. N., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., & Sigmond, M. (2019j). CCCma CanESM5 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3696
- Tachiiri, K., Abe, M., Hajima, T., Arakawa, O., Suzuki, T., Komuro, Y., & Kawamiya, M. (2019a). MIROC MIROC-ES2L model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5742 Tachiiri, K., Abe, M., Hajima, T., Arakawa, O., Suzuki, T., Komuro, Y., & Kawamiya, M. (2019b). MIROC MIROC-ES2L model output pre-
- pared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5745 Tachiiri, K., Abe, M., Hajima, T., Arakawa, O., Suzuki, T., Komuro, Y., & Kawamiya, M. (2019c). MIROC MIROC-ES2L model output pre-
- pared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5751 Tachiiri, K., Abe, M., Hajima, T., Arakawa, O., Suzuki, T., Komuro, Y., & Kawamiya, M. (2019d). MIROC MIROC-ES2L model output pre-
- pared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5770 Tang, Y., Rumbold, S., Ellis, R., Kelley, D., Mulcahy, J., Sellar, A., & Jones, C. (2019). MOHC UKESM1.0-LL model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6113
- Tatebe, H., & Watanabe, M. (2018). MIROC MIROC6 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5603

- Tedesco, M., & Fettweis, X. (2020a). Unprecedented atmospheric conditions (1948–2019) drive the 2019 exceptional melting season over the Greenland ice sheet. *The Cryosphere*, 14(4), 1209–1223. https://doi.org/10.5194/tc-14-1209-2020
- Tedesco, M., & Fettweis, X. (2020b). Unprecedented atmospheric conditions (1948–2019) drive the 2019 exceptional melting season over the Greenland ice sheet. *The Cryosphere*, 14(4), 1209–1223. https://doi.org/10.5194/tc-14-1209-2020
- van Angelen, J. H., Lenaerts, M. J. T., van den Broeke, M. R., Fettweis, X., & van Meijgaard, E. (2013). Rapid loss of firn pore space accelerates 21st century Greenland mass loss. *Geophysical Research Letters*, 40, 2109–2113. https://doi.org/10.1002/grl.50490
- van den Broeke, M. R., Enderlin, E. M., Howat, I. M., Kuipers Munneke, P., Noël, B. P. Y., van de Berg, W. J., et al. (2016). On the recent contribution of the Greenland ice sheet to sea level change. *The Cryosphere*, *10*(5), 1933–1946. https://doi.org/10.5194/tc-10-1933-2016
- van Kampenhout, L., Lenaerts, J. T. M., Lipscomb, W. H., Lhermitte, S., Noël, B., Vizcaíno, M., et al. (2020). Present-Day Greenland ice sheet climate and surface mass balance in CESM2. Journal of Geophysical Research: Earth Surface, 125, e2019JF005318. https://doi. org/10.1029/2019JF005318
- van Kampenhout, L., Lenaerts, J. T. M., Lipscomb, W. H., Sacks, W. J., Lawrence, D. M., Slater, A. G., & van den Broeke, M. R. (2017). Improving the representation of polar snow and firm in the Community Earth System Model. *Journal of Advances in Modeling Earth Systems*, 9, 2583–2600. https://doi.org/10.1002/2017MS000988
- Vizcaino, M., Lipscomb, W. H., Sacks, W. J., & van den Broeke, M. (2014). Greenland surface mass balance as simulated by the Community Earth System Model. Part II: Twenty-First-Century changes. *Journal of Climate*, 27(1), 215–226. https://doi.org/10.1175/ JCLI-D-12-00588.1
- Voldoire, A. (2018). CMIP6 simulations of the CNRM-CERFACS based on CNRM-CM6-1 model for CMIP experiment historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4066

Voldoire, A. (2019a). CNRM-CERFACS CNRM-CM6-1-HR model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4067

Voldoire, A. (2019b). CNRM-CERFACS CNRM-CM6-1-HR model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4190

- Voldoire, A. (2019c). CNRM-CERFACS CNRM-CM6-1-HR model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4225
- Voldoire, A. (2019d). CNRM-CERFACS CNRM-CM6-1 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4184
- Voldoire, A. (2019e). CNRM-CERFACS CNRM-CM6-1 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4189
- Voldoire, A. (2019f). CNRM-CERFACS CNRM-CM6-1 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4197

Voldoire, A. (2019g). CNRM-CERFACS CNRM-CM6-1 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4224

Voldoire, A. (2019h). CNRM-CERFACS CNRM-ESM2-1 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4186

Voldoire, A. (2019i). CNRM-CERFACS CNRM-ESM2-1 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4191

- Voldoire, A. (2019j). CNRM-CERFACS CNRM-ESM2-1 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4199
- Voldoire, A. (2019k). CNRM-CERFACS CNRM-ESM2-1 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4226
- Voldoire, A. (2020a). CNRM-CERFACS CNRM-CM6-1-HR model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4185

Voldoire, A. (2020b). CNRM-CERFACS CNRM-CM6-1-HR model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4198

Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019a). INM INM-CM4-8 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5069

- Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019b). INM INM-CM4-8 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.12325
- Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019c). INM INM-CM4-8 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.12327
- Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019d). INM INM-CM4-8 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.12329
- Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019e). INM INM-CM4-8 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.12337
 Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019f). INM INM-CM5-0 model output prepared

for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.5070

- Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019g). INM INM-CM5-0 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.12326 Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019h). INM INM-CM5-0 model output prepared
- for CMIP6 Scenario MIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.2203/ESGF/CMIP6.12328 Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019i). *INM INM-CM5-0 model output prepared*
- for CMIP Scenario MIP ssp370 (Versio 2020512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.12330 Volodin, E., Mortikov, E., Gritsun, A., Lykossov, V., Galin, V., Diansky, N., & Emelina, S. (2019j). *INM INM-CM5-0 model output prepared*
- for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.12338 Wake, L., & Marshall, S. (2015). Assessment of current methods of positive degree-day calculation using in situ observations from glaciated
- wake, L., & Marshall, S. (2015). Assessment of current methods of positive degree-day calculation using in situ observations from glaciated regions. *Journal of Glacology*, *61*(226), 329–344. https://doi.org/10.3189/2015JoG14J116
- Wang, W., Zender, C. S., van As, D., & Miller, N. B. (2019). Spatial distribution of melt season cloud radiative effects over Greenland: Evaluating satellite observations, reanalyses, and model simulations against in situ measurements. *Journal of Geophysical Research:* Atmospheres, 124, 57–71. https://doi.org/10.1029/2018JD028919
- Wieners, K.-H., Giorgetta, M., Jungclaus, J., Reick, C., Esch, M., Bittner, M., & Roeckner, E. (2019a). MPI-M MPI-ESM1.2-LR model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6595



 Wieners, K.-H., Giorgetta, M., Jungclaus, J., Reick, C., Esch, M., Bittner, M., & Roeckner, E. (2019b). MPI-M MPI-ESM1.2-LR model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6690
 Wieners, K.-H., Giorgetta, M., Jungclaus, J., Reick, C., Esch, M., Bittner, M., & Roeckner, E. (2019c). MPI-M MPI-ESM1.2-LR model output

prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6693 Wieners, K.-H., Giorgetta, M., Jungclaus, J., Reick, C., Esch, M., Bittner, M., & Roeckner, E. (2019d). MPI-M MPI-ESM1.2-LR model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6695

Wieners, K.-H., Giorgetta, M., Jungclaus, J., Reick, C., Esch, M., Bittner, M., & Roeckner, E. (2019e). MPI-M MPI-ESM1.2-LR model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6705
 Wu, T., Chu, M., Dong, M., Fang, Y., Jie, W., Li, J., & Zhang, Y. (2018). BCC BCC-CSM2MR model output prepared for CMIP6 CMIP historical

(Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.2948 Xin, X., Wu, T., Shi, X., Zhang, F., Li, J., Chu, M., & Wei, M. (2019a). BCC BCC-CSM2MR model output prepared for CMIP6 ScenarioMIP

- ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3028 Xin, X., Wu, T., Shi, X., Zhang, F., Li, J., Chu, M., & Wei, M. (2019b). BCC BCC-CSM2MR model output prepared for CMIP6 ScenarioMIP
- ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3030
 Xin, X., Wu, T., Shi, X., Zhang, F., Li, J., Chu, M., & Wei, M. (2019c). BCC BCC-CSM2MR model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3035

Xin, X., Wu, T., Shi, X., Zhang, F., Li, J., Chu, M., & Wei, M. (2019d). BCC BCC-CSM2MR model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3050

Xu, B., Wang, N., Chen, T., & Li, M. (2015). Empirical evaluation of rectified activations in convolutional network. Cornell University.

Yukimoto, S., Koshiro, T., Kawai, H., Oshima, N., Yoshida, K., Urakawa, S., & Adachi, Y. (2019a). MRI MRI-ESM2.0 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6842

Yukimoto, S., Koshiro, T., Kawai, H., Oshima, N., Yoshida, K., Urakawa, S., & Adachi, Y. (2019b). MRI MRI-ESM2.0 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6909

Yukimoto, S., Koshiro, T., Kawai, H., Oshima, N., Yoshida, K., Urakawa, S., & Adachi, Y. (2019c). MRI MRI-ESM2.0 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6910

Yukimoto, S., Koshiro, T., Kawai, H., Oshima, N., Yoshida, K., Urakawa, S., & Adachi, Y. (2019d). MRI MRI-ESM2.0 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6915

Yukimoto, S., Koshiro, T., Kawai, H., Oshima, N., Yoshida, K., Urakawa, S., & Adachi, Y. (2019e). MRI MRI-ESM2.0 model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.6929

Yu, Y. (2019a). CAS FGOALS-f3-L model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3355

Yu, Y. (2019b). CAS FGOALS-f3-L model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3464

Yu, Y. (2019c). CAS FGOALS-f3-L model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3468

Yu, Y. (2019d). CAS FGOALS-f3-L model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3479

Yu, Y. (2019e). CAS FGOALS-f3-L model output prepared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.3502

Zelinka, M. D., Myers, T. A., McCoy, D. T., Po-Chedley, S., Caldwell, P. M., Ceppi, P., et al. (2020). Causes of higher climate sensitivity in CMIP6 Models. *Geophysical Research Letters*, 47, e2019GL085782. https://doi.org/10.1029/2019GL085782

Zhang, J., Wu, T., Shi, X., Zhang, F., Li, J., Chu, M., & Wei, M. (2018). BCC BCC-ESM1 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.2949

Ziehn, T., Chamberlain, M., Lenton, A., Law, R., Bodman, R., Dix, M., & Druken, K. (2019a). CSIRO ACCESS-ESM1.5 model output prepared for CMIP6 CMIP historical (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4272

Ziehn, T., Chamberlain, M., Lenton, A., Law, R., Bodman, R., Dix, M., & Druken, K. (2019b). CSIRO ACCESS-ESM1.5 model output prepared for CMIP6 ScenarioMIP ssp126 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4320

Ziehn, T., Chamberlain, M., Lenton, A., Law, R., Bodman, R., Dix, M., & Druken, K. (2019c). CSIRO ACCESS-ESM1.5 model output prepared for CMIP6 ScenarioMIP ssp245 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4322 Ziehn, T., Chamberlain, M., Lenton, A., Law, R., Bodman, R., Dix, M., & Druken, K. (2019d). CSIRO ACCESS-ESM1.5 model output pretransport of the statement of the statemen

Ziehn, T., Chamberlain, M., Lenton, A., Law, R., Bodman, R., Dix, M., & Druken, K. (2019d). CSIRO ACCESS-ESM1.5 model output prepared for CMIP6 ScenarioMIP ssp370 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4324 Ziehn, T., Chamberlain, M., Lenton, A., Law, R., Bodman, R., Dix, M., & Druken, K. (2019e). CSIRO ACCESS-ESM1.5 model output pre-

pared for CMIP6 ScenarioMIP ssp585 (Version 20200512). Earth System Grid Federation. https://doi.org/10.22033/ESGF/CMIP6.4333